2022, 21, Dow



Geophysical Research Letters[®]

RESEARCH LETTER

10.1029/2021GL096337

Key Points:

- The inclusion of topography in an idealized model setup yields global-scale slow-moving Rossby waves with preferred longitudinal locations
- When these slow-moving Rossby waves amplify they lead to an increased frequency of concurrent heatwaves in preferred longitudinal locations
- In the model, zonal wavenumbers 5 and 6 exhibit the strongest anomalies, which strongly impact temperature and heatwave frequency

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

B. Jiménez-Esteve, bernat.jimenez@env.ethz.ch

Citation:

Jiménez-Esteve, B., Kornhuber, K., & Domeisen, D. I. V. (2022). Heat extremes driven by amplification of phase-locked circumglobal waves forced by topography in an idealized atmospheric model. *Geophysical Research Letters*, *49*, e2021GL096337. https://doi. org/10.1029/2021GL096337

Received 27 SEP 2021 Accepted 21 OCT 2022

Author Contributions:

Conceptualization: B. Jiménez-Esteve, K. Kornhuber, D. I. V. Domeisen Formal analysis: B. Jiménez-Esteve Funding acquisition: D. I. V. Domeisen Investigation: B. Jiménez-Esteve, K. Kornhuber Methodology: B. Jiménez-Esteve Supervision: D. I. V. Domeisen Validation: B. Jiménez-Esteve Visualization: B. Jiménez-Esteve Writing – original draft: B. Jiménez-Esteve

© 2022. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use,

distribution and reproduction in any medium, provided the original work is properly cited.

Heat Extremes Driven by Amplification of Phase-Locked Circumglobal Waves Forced by Topography in an Idealized Atmospheric Model

B. Jiménez-Esteve¹, K. Kornhuber², and D. I. V. Domeisen^{1,3}

¹ETH Zürich, Institute for Atmospheric and Climate Science, Zürich, Switzerland, ²Earth Institute, Columbia University, New York, NY, USA, ³Institute of Earth Surface Dynamics, University of Lausanne, Lausanne, Switzerland

Abstract Heatwaves are persistent temperature extremes associated with devastating impacts on human societies and ecosystems. In the midlatitudes, amplified quasi-stationary Rossby waves have been identified as a key mechanism for heatwave occurrence. Amplified waves with preferred longitudinal locations lead to concurrent extremes in specific locations. It is therefore important to identify the essential components in the climate system that contribute to phase-locking of wave patterns. Here, we investigate the role of dry atmospheric dynamics and topography in causing concurrent heatwaves by using an idealized general circulation model. Topography is included in the model experiments as a Gaussian mountain. Our results show that amplified Rossby waves exhibit clear phase-locking behavior and a decrease in the zonal phase speed when a large-scale localized topographic forcing is imposed, leading to concurrent heat extremes at preferred locations.

Plain Language Summary Heatwaves are among the most deadly natural disasters. Since climate projections indicate an increase in the severity and frequency of heatwaves, it is important to understand which processes are responsible for their occurrence. Heatwaves in midlatitudes tend to be associated with atmospheric waves, characterized by a strong meandering of the jet stream. Occasionally, waves tend to reduce their longitudinal propagation speed along with an amplification of their amplitude in preferred locations. This behavior is known as phase-locking and can lead to extreme weather events, occurring simultaneously in different regions, as for example, during the Russian heatwave and Pakistan flooding in 2010. Here, we investigate the role of large-scale mountains in driving concurrent heat extremes. We use simulations from an idealized atmospheric model with an idealized mountain, and keeping only the essential atmosphere dynamics represented. In the model, topography alone leads to the occurrence of phase-locked circumglobal waves and localized heatwaves. The eastward velocity of the waves is substantially reduced due to topography, which can be linked to an increase in heatwaves in specific areas. This research lays the basis for understanding the fundamental factors that lead to concurrent extreme weather events, which is important to improve their forecasts.

1. Introduction

Heatwaves remain one of the deadliest geophysical events (Hoegh-Guldberg et al., 2018; Koppe et al., 2004; Yaghmaei, 2020) and are projected to increase in magnitude and frequency with climate change (Lopez et al., 2018; Perkins et al., 2012; Perkins-Kirkpatrick & Lewis, 2020; Watanabe et al., 2013). A thorough understanding of the physical driving mechanisms is necessary to improve their prediction and to identify regions at risk under future emission scenarios. High impact extreme weather events in the Northern Hemisphere midlatitudes can be associated with slow moving, that is, quasi-stationary, and high-amplitude Rossby waves, which can remain in place for several days to weeks (Petoukhov et al., 2013). Most notable examples include the European heatwaves of 2003 (Kornhuber, Petoukhov, Petri, et al., 2017), 2015 (Duchez et al., 2016) and 2018 (Drouard et al., 2019; Kornhuber et al., 2019; Vogel et al., 2019; Wehrli et al., 2020), the Russian heatwave of 2010 (Di Capua et al., 2021; Lau & Kim, 2012), and the recent North American heatwave of 2021 (Philip et al., 2021). Such extreme weather events often do not occur in isolation, but are a regional imprint of a hemispheric pattern with extreme heat and rainfall extremes occurring almost simultaneously in different longitudinal locations (Coumou et al., 2014; Kornhuber et al., 2020). For instance, the circumglobal wave pattern that contributed to the 2018 European heatwave also promoted extreme rainfall and flooding in Greece, a heatwave in the Caspian Sea region and extreme rainfall

1 of 12

ᠭ



Writing – review & editing: B. Jiménez-Esteve, K. Kornhuber, D. I. V. Domeisen in Japan (Kornhuber et al., 2019), as well as a concurrent heatwave in California (Domeisen et al., 2022). The amplified wave pattern that fueled the extreme heat over northwestern North America in late June 2021 occurred almost simultaneously with an extreme heatwave in Scandinavia and Siberia.

Topography and thermal contrast, for example, across land—ocean boundaries, act on the atmospheric circulation and favor particular wave patterns of regional and hemispheric extent that exhibit a strong seasonal cycle (Chang, 2009; Charney & Eliassen, 1949; Held et al., 2002). For example, the Rocky Mountains force a stationary ridge over the North American continent (e.g., Brayshaw et al., 2009; Saulière et al., 2012), which might have strengthened in recent winters (Singh et al., 2016). Similarly, the subtropical jet over Eurasia is influenced by the Tibetan plateau and Mongolian mountains (White et al., 2017) and exhibits a recurrent flow pattern called the Silk road pattern, which is associated with co-variability ranging from Europe to east Asia (Boers et al., 2019; Ding & Wang, 2005; Kosaka et al., 2009; Lu et al., 2002). Given sufficiently strong forcing and suitable background conditions (also known as "waveguidability," see Manola et al. (2013); Wirth (2020); Wirth and Polster (2021); White et al. (2021)), Rossby waves can form circumglobal teleconnections (Branstator, 2002; Teng & Branstator, 2019). As the location and strength of the background flow changes seasonally, different circumglobal teleconnection patterns have been identified in summer and winter (Branstator, 2002; Ding & Wang, 2005; Kornhuber et al., 2020). In summer, recurrent wave 5 and 7 patterns have been identified that cause concurrent heatwaves in specific regions in the Northern Hemisphere midlatitudes (Kornhuber et al., 2020). These wave patterns have been detected during some of the most extreme heat events of recent years: The European heatwaves of 2003 and 2018 have been diagnosed as a wave 7 pattern (Kornhuber et al., 2019), while the 2010 heatwave was associated with a wave 5-6 pattern (Figure 2 in Petoukhov et al. (2013) and Di Capua et al. (2021)).

Several mechanisms have been suggested to cause such amplified wave patterns. Quasi-resonant amplification of planetary waves (Petoukhov et al., 2013) describes the process during which a waveguide forms under specific zonal mean flow conditions (Manola et al., 2013), which allows waves to amplify given sufficient forcing. Studies investigating the amplification of circumglobal wave patterns and phase-locking behavior, that is, the fact that under stationary forcing waves of certain wavenumbers exhibit preferred longitudinal locations, so far are based on reanalysis data (e.g., Kornhuber, Petoukhov, Karoly, et al., 2017; Kornhuber, Petoukhov, Petri, et al., 2017; Petoukhov et al., 2018) or more complex atmospheric general circulation models (Huntingford et al., 2019), for which it can be challenging to isolate specific dynamical mechanisms. Thus the analysis of wave amplification and phase-locking of amplified waves under more idealized conditions is still needed for process understanding (White et al., 2022). Here, we investigate the role of topographic forcing in idealized model experiments, where only basic atmospheric dynamics are represented together with topography. This model setup allows us to focus on how localized topography affects the propagation of midlatitude Rossby waves and influences the preferred location where troughs and ridges associated with circumglobal patterns amplify and lead to an increased like-lihood of temperature extremes. This approach further clarifies the interaction between orographically forced Rossby waves and transient Rossby waves originating from baroclinic instability.

2. Data and Methods

2.1. Model Setup

We use an idealized configuration of the ICOsahedral Nonhydrostatic (ICON) atmospheric model (Zängl et al., 2015) on a R_2B_4 triangular grid, with an equivalent ~158 km grid spacing, and 41 vertical levels. We turn off all the physical parameterizations and substitute them by a simple zonal mean Newtonian temperature relaxation parameterization as in Held and Suarez (1994). A more detailed description of the model setup can be found in Jiménez-Esteve and Domeisen (2022).

Idealized topography is introduced in the form of a 2-dimensional Gaussian mountain with a maximum height H at a latitude and longitude of $(\phi_0, \lambda_0 = 90^{\circ}\text{E})$. The half-widths of the mountain are fixed to $\sigma_x = \sigma_y = 1,500$ km. The mathematical formulation for the topography follows Cook and Held (1992), except that in our case the half-widths are given in km (instead of °), which preserves the volume of the mountain when changing its latitude. We use three topography configurations: $(H, \phi_0) = (4 \text{ km}, 45^{\circ}\text{N}), (H, \phi_0) = (8 \text{ km}, 45^{\circ}\text{N}), \text{ and } (H, \phi_0) = (8 \text{ km}, 25^{\circ}\text{N})$. Each simulation is run for a total of 30 years (360 days a year) using the same equilibrium temperature and relaxation timescale as in Held and Suarez (1994).

2022,

The zonal wind climatology at 300 hPa (u300) of the model simulations and ERA-Interim reanalysis (Dee et al., 2011) are shown in Figure S1 in Supporting Information S1. It is important to note that our model does not simulate a seasonal cycle, and background conditions are close to the observed annual mean conditions.

2.2. Zonal Wavenumber Decomposition, Phase-Locking, and Hayashi Spectra

We use the meridional wind at 300 hPa (v300) to identify Rossby waves in the midlatitude troposphere. In an idealized framework a Rossby wave can be mathematically written as a perfect zonal sinusoidal circumglobal wave:

$$v300(\lambda, t) = A(t)\cos(k\lambda + \Phi(t))$$
(1)

where λ is the longitude in radians, t the time, A(t) > 0 is the amplitude of the wave, k = 1, 2, 3, ... the non-dimensional zonal wavenumber, $\Phi(t)$ is the phase of the wave in radians at time t.

At every given time *t* the v300 field can be decomposed into a range of superimposed cosine waves of different wavenumbers *k*, with amplitudes A_k and phases Φ_k . We first calculate the area-weighted latitudinal mean between 60° and 30°N and then apply a 7-day running mean to the v300 field in order to filter out medium- and high-frequency transient waves. Then, a Fast Fourier Transformation in the longitude dimension is applied to obtain the amplitude (A_k) and the phase (Φ_k) for each wavenumber *k*. It is important to highlight that for our wave amplitude and phase calculation we do not subtract the model climatology. Temporal wave anomalies cannot have a phase preference as by definition the time average must be zero at all points. The climatology and percentiles of the obtained amplitudes for each simulation and reanalysis can be found in Figure S2 in Supporting Information S1 for reference. We focus on wavenumbers k = 4-8, since synoptic-scale waves are commonly involved in circumglobal wave amplification events, for example, k = 4 (Petoukhov et al., 2018), k = 5 (Harnik et al., 2016; Teng et al., 2013), or k = 6-8 (Mann et al., 2017; Petoukhov et al., 2013, 2016), and exhibit phase locking behavior in reanalysis (Kornhuber, Petoukhov, Petri, et al., 2017).

To study the high-amplitude episodes that are characterized by single (phase-locked) wavenumbers, we calculate composites by averaging the unfiltered v300 anomalies, temperature at 1,000 hPa (T1000) anomalies (with respect to the same model run) and heatwave frequency anomalies (see Section 2.3 for definition) over time steps when $A_k > 1.5$ standard deviations (SDs) for each specific wavenumber k. To analyze the temporal evolution of high-amplitude events, we define the onset of a wave amplification event as the first day of at least three consecutive days of $A_k > 1.5SD$ for each wavenumber k.

To quantify phase-locking we define the index δ_m as the narrowest window that contains half of the probability density of the phase distribution. Lower values of δ_m correspond to stronger phase-locking. To estimate the phase speed *c* of the waves, we compute the Hayashi spectrum (Hayashi, 1979) using the methodology by Randel and Held (1991). For an application of this diagnostic see Domeisen et al. (2018) and Riboldi et al. (2021). A more detailed description of these two methodologies can be found in Supporting Information S1. In this study we use the term "quasi-stationary" waves to refer to wave anomalies that are slow moving, that is, their phase-speed is close to zero, while the term "stationary" waves is used to refer to the climatological waves.

2.3. Heatwave Definition

Heatwaves are identified as periods of three or more consecutive extreme heat days, which are defined as days when the temperature at the 1,000 hPa pressure level (T1000) is above the 95th percentile. Other pressure levels close to the surface yield comparable results. The 95th percentile is computed independently for each grid point and for each model experiment using unfiltered data. We do not consider a minimum temporal separation between events. The heatwave frequency is defined as the fraction of days that belong to heatwaves.

3. Results

To quantify which wavenumbers exhibit stronger phase-locking when topography is included in the model, we look at the 2-dimensional distribution of the amplitude versus phase for each zonal wavenumber (Figure 1). See Section 2.2 for details on the calculation. For the simulation without topography (no-topo) (Figures 1a-1e) zonal



Geophysical Research Letters



Figure 1. Two-dimensional histograms of wave amplitude versus phase of the meridional wind at 300 hPa (v300) for the model experiment (a–e) without topography (no-topo), (f–j) with 4 km topography centered at (45°N, 90°E), (k–o) with 8 km topography centered at (45°N, 90°E), and (p–t) 8 km topography centered at (25°N, 90°E). Each column shows a different zonal wavenumber (k = 4–8). The mean and the 1.5 SD amplitude threshold are indicated as horizontal red lines. The vertical dashed line indicates the phase of maximum probability for all amplitudes. The phase-locking index (δ_m in radians) is displayed in the upper-right corner of each panel, also for the high-amplitude cases (a lower δ_m indicates a stronger phase-locking for a particular zonal wavenumber). A total of 10,794 values (30 years of 360 days each, where the first and last 3 days are disregarded due to the 7-day running mean window) are used to produce each of the histograms. No smoothing is applied.

wavenumbers k = 4-8 indicate no preference for phase-locking. In other words, when waves amplify, that is, when their wave amplitude increases to above average values, their maxima can occur with equal probability in any longitudinal location. In contrast, in the experiments with an 8 km mountain located in midlatitudes (45°N, Figures 1k–1o) or in the subtropics (25°N, Figures 1p–1t) waves are more likely to amplify for a specific phase, that is, longitudinal location, meaning that they are phase-locked. The same effect but of weaker intensity can be observed for the smaller 4-km mountain (Figures 1f–1j). Phase-locking does not occur with the same amplitude for the different synoptic wavenumbers, and it is particularly strong for wavenumbers 4, 5 and 6 (see lower values of δ_m in Figure 1 and Figure S3 in Supporting Information S1). Large amplitude events ($A_k > 1.5SD$) tend to exhibit higher values of phase-locking for extreme amplitude events compared to climatology, consistent with results found by Kornhuber et al. (2020) using reanalysis data. For the remainder of this study we will predominantly focus on wavenumbers 5 and 6, as those wavenumbers exhibit higher amplitudes than wavenumber 4 in our model configuration, while other wavenumbers are discussed in less detail. Larger synoptic wavenumbers (k > 6)





Figure 2. Hayashi spectra of v300 anomalies (m/s) as a function of phase speed (m/s) relative to the ground and zonal wavenumber k for (a) the no-topography simulation (shading) and (b–d) three different topographic forcings (black contours). The difference between the spectral densities of each topography experiment and the no-topo experiment is shown as color shading in panels (b–d). The black line in panel (a) indicates a constant period of 7 days for reference.

usually exhibit weaker phase locking and maximum amplitudes and thus their impact on extreme heat events is expected to be less important. Note that the higher amplitudes for waves 5 and 6 can also be observed for the no-topo experiment (Figures 1b and 1c, Figure S2 in Supporting Information S1), and is probably a consequence of baroclinic instability in our model configuration (Grotjahn, 2015). Note that there is no asymmetric forcing that can force waves in the no-topo run and thus the largest amplitudes found for wavenumbers 5 and 6 are independent of topographic forcing.

As a next step, we examine how topography impacts the phase speed of waves. The Hayashi spectrum (see Section 2.2 and Supporting Information S1) allows us to quantify the distribution of the power spectrum for the midlatitude flow in terms of zonal wavenumber and phase speed (Figure 2). Note that we do not apply any temporal filter to the data in order to represent the full spectrum of waves. For the no-topo run the spectral density is maximized for synoptic wavenumbers k = 4-7 with a peak around k = 5, 6 (Figure 2a), consistent with the maximum amplitudes observed in Figure 1 and Figure S2 in Supporting Information S1. A similar spectrum is found for reanalysis during equinoxes (Riboldi et al., 2021), which highlights the ability of our model to simulate realistic atmospheric conditions. Figure 2a also shows how larger wavenumbers tend to be associated with faster eastward propagation, which is a consequence of the Rossby wave dispersion relation ($c = U - \beta / (k^2 + l^2)$). Thus, small wavenumbers (k < 7) can more often become quasi-stationary (slow-moving) if the mean flow in the midlatitude troposphere slows down sufficiently.

We now investigate the effect of localized topographic forcing on the Hayashi spectra (Figures 2b–2d). Already a moderate forcing (4 km, 45°N) has an effect on the spectra, with an increasing power of quasi-stationary waves,





Figure 3. Mean composites of (a) anomalies of meridional wind at 300 hPa (v300), (b) anomalies of air temperature at 1,000 hPa (T1000), and (c) heatwave frequency relative anomaly (note the nonlinear colorbar [-100 to 0 in 20% intervals; 0 to 200 in 50% intervals]) for high-amplitude (A > 1.5 SD) days of zonal wavenumbers k = 4-7 (top to bottom) identified in the 8 km and 25°N topography experiment. In each panel, the geopotential height at 300 hPa is displayed as black contours (8,800, 9,000, and 9,200 m) and topographic elevation in purple contours (1, 3, and 5 km). Note that geopotential height contours at smaller distances correspond to a stronger geostrophic wind. The heatwave frequency relative anomaly shown in panel (c) is the percentage change in heatwave frequency relative to the climatological frequency of heatwaves in each model simulation. Only statistically significant anomalies above the 95% confidence level are shown according to a two-tailed *t*-test.

especially for wavenumbers 6–8, and decreasing eastward propagating synoptic waves with respect to the no-topo run (Figure 2b). This effect is stronger for a doubled mountain height located at the same latitude (8 km, 45°N), with a clear decrease in the eastward phase speed spectral density (Figure 2c). For a subtropical topography forcing (8 km, 25°N) there is also a decrease in eastward propagating synoptic waves, while slow-moving westward propagating waves are more common than in the no-topography simulation (Figure 2d).

To analyze the impact on near-surface temperature and persistent heat extremes associated with amplification of phase-locked waves we show composites of the unfiltered v300, T1000 and the heatwave frequency relative anomaly for all days when a high amplitude of a specific wavenumber is identified in the (8 km, 25°N) topography simulation (Figure 3). Note that the waves shown in Figure 3 are anomalies with respect to the model's zonally varying climatology, that is, they are in addition to the stationary waves excited by topography. Qualitatively similar results are obtained for the midlatitude forcings (Figures S4 and S5 in Supporting Information S1), although we find that the response for the subtropical mountain exhibits a more pronounced wave pattern of near-hemispheric extent. Figure 3 confirms the circumglobal and phase-locked character of the high-amplitude events identified in our simulations, which can be seen as a series of wind anomalies of alternating sign extending to the full longitude circle in midlatitudes (Figure 3a). Note that the circumglobal characteristic for wavenumbers

5 and 6 is not just a result of the multi-event average (see Figure S8 in Supporting Information S1). The same analysis with the no-topo run (not shown) leads to negligible anomalies, as although the maximum amplitudes can be similar they do not occur with a preferred phase (Figures 1a–1e). The composited anomalies have approximately twice the magnitude for k = 5 and 6 as compared to k = 4 and 7, consistent with Figures 1 and 2. Associated with the upper level troughs and ridges (black contours in Figure 3) T1000 anomalies show a circumglobal pattern, which is approximately 90° out of phase with the v300 anomaly field. Interestingly, T1000 and v300 anomalies are stronger upstream than downstream of the topography (less clear for k = 7). This weaker response downstream is likely a consequence of a stronger zonal jet speed (Figure S1 in Supporting Information S1), which is related to the stronger meridional temperature gradients downstream of large-scale topography (e.g., Lutsko et al., 2019).

The effect of circumglobal wave amplification on heatwave occurrence is also shown in Figure 3c. Consistent with the temperature anomalies near the surface, heatwave frequency increases in the regions with atmospheric ridges, associated with a relative increase of up to 300% mainly for wavenumbers 5 and 6. This means a heatwave is four times more likely to occur during a wave amplification event than for climatological conditions in this specific longitudinal location. In locations where troughs amplify we observe a clear decrease in the probability of having a heatwave of up to a 100%.

To gain a deeper insight into the dynamical evolution of wave amplification events, we compute the multi-event average of v300 and T1000 with respect to the onset date (see Section 2.2). Figure 4 displays the temporal evolution of the v300 and T1000 anomalies averaged over 30° - 60° N for amplification events of k = 5 and 6 in the three model simulations. We display the results for wavenumbers 5 and 6 as those show the strongest amplitudes and more phase-locking in our model simulations (Figure 1). The results for k = 4,7 are shown in Figure S6 in Supporting Information S1. The amplitudes of wavenumber 5 and 6 strongly dominate the signal during amplification events, while this is not the case for the amplification of other zonal wavenumbers, as their amplitudes are in general much weaker than the amplitudes of the dominant wavenumbers (k = 5, 6) (see Figure S8 in Supporting Information S1).

The Hovmöller diagram (Figure 4) shows that wave amplification events tend to last on average 6–7 days for both k = 5 and 6 (see also Figure S9 in Supporting Information S1). The quasi-stationary wave anomalies (with respect to the zonally varying model climatology) show a similar phasing as the climatological stationary waves generated by topography (Figure S7 in Supporting Information S1). It is interesting to note the westward phase speed for the disturbances located upstream of the mountain peak (mainly for k = 5 in the TOPO [8 km, 45°N] experiment). This westward phase speed is likely due to slower zonal winds upstream than downstream of the mountain (Figure S1 in Supporting Information S1) and due to the Rossby wave dispersion relation, which states that wavenumber 5 propagates faster westward with respect to the mean flow than wavenumber 6. Note that the stronger v300 anomalies in the averaged composite for TOPO (8 km, 45°) compared to TOPO (4 km, 45°) result from a more intense phase-locking for the stronger forcing (Figure 1). When the topography is located south of the jet axis (Figures 4e and 4f), v300 anomalies associated with k = 5, 6 wave amplification tend to be more circumglobal (a similar magnitude for all longitudes) than for the midlatitude mountain. The phase speed (tilt of the v300 anomalies) during these events tends to be close to zero, especially for k = 5 and upstream of the mountain, where zonal winds tend to be weaker (Figure S1 in Supporting Information S1).

Another important question is whether there are clear precursors to wave amplification events. Rossby wave packets (Wirth et al., 2018) of limited longitudinal extent develop up to 15–20 days before the onset of some wave amplification events, with energy propagating eastwards (positive group velocity). These precursors are clearer in some of the composites, see for example, Figures 4a and 4d (between 90°W and 90°E and 0–10 days before the onset). For k = 6 and a subtropical mountain (Figure 4f) v300 anomalies emerge even earlier and amplify 15 days before the onset near the location of the mountain before the initiation of the circumglobal response. Future work should investigate the relationship between the Rossby wave packets and the circumglobal wave amplification.

4. Discussion and Conclusions

In this study, we show evidence that localized topography leads to phase-locked amplification of quasi-stationary synoptic-scale waves. We base our analysis on idealized simulations using only thermal forcing and Rayleigh surface drag as in (Held & Suarez, 1994), with no other physical parameterizations, which allows us to demonstrate that very few "ingredients" are needed to produce phase-locked and amplified waves and associated concurrent



Geophysical Research Letters



Figure 4. Hovmöller diagram (time vs. longitude) of daily mean v300 anomalies (shading) and T1000 anomalies (blue contours: from -2.5 to 2.5 in 1 K intervals, negative dashed) averaged over 30° – 60° N and composited with respect to the onset of high-amplitude (a, c, d) k = 5 and (b, d, f) k = 6 events. (a and b) 4 km mountain at 45°N, (c and d) 8 km mountain at 45°N, and (e and f) 8 km mountain at 25°N. Only statistically significant v300 anomalies above the 95% confidence level are shown according to a two-tailed *t*-test.

heatwaves. Observed phase-locking events have been shown to be associated with simultaneous extreme events over major crop regions across the globe (e.g., Kornhuber et al., 2020). Our results confirm that a similar connection between heatwaves and phase-locked amplified waves exists in a simple model configuration, originating solely from topographic forcing and dry atmospheric dynamics.

These experiments highlight that changes in the relative position of the mean flow with respect to the topographic disturbance can lead to different responses in terms of the amplitude of circumglobal phase-locked waves, which

impacts the frequency and location of heatwave hotspots (cf., Figure 3 and Figure S3 in Supporting Information S1). Observations show a weak poleward shift of the midlatitude jet (Osman et al., 2021) that is projected to continue under high emission warming scenarios (Harvey et al., 2020). This shift results in a different location of the jet with respect to the topographic forcing, although this relative shift between the jet and the topography is considerably smaller on average than in our idealized simulations. Nevertheless, our results suggest that a jet shift might lead to a change in future heatwave hotspots. While past discussions on atmospheric circulation changes and impacts on summer weather extremes have put a primary focus on changes in the jet speed (Coumou et al., 2015; Kornhuber & Tamarin-Brodsky, 2021) and its ability to support resonant waves (Mann et al., 2018), our results suggest that changes in jet latitude relative to important mountains should be given similar attention. Apart from the location, the shape and height of the topographic forcing can impact the stationary wave response (Brayshaw et al., 2009; Valdes & Hoskins, 1991). In this study, we keep the shape of the mountain fixed to a half-width of 1,500 km, which is comparable to the scale of the Tibetan plateau or the Rocky mountains (see Figure S1 in Supporting Information S1). Moreover, nonlinear interactions with other mountain ranges (Narinesingh et al., 2020) and diabatic forcing originating from land-sea contrast and tropical convection (Garfinkel et al., 2020; Kaspi & Schneider, 2011) modulate the total stationary wave response. Using a simplified version of topography allows us to clearly identify the main role of topographic forcing.

While this analysis focuses on the relationship between topographic forcing and phase-locking, the forcing leading to circumglobal teleconnections can be more diverse under real-world conditions (Branstator, 2002; Teng & Branstator, 2019). Apart from the topographic forcing investigated here, thermal forcing, for example, from land-sea thermal contrast, can be responsible for preferred positions of Rossby waves. In addition, phase-locked wave trains can arise from SST anomalies (Beverley et al., 2021) or dry soil conditions over large continental areas (Teng et al., 2019).

Our work relies on a constant thermal forcing throughout the experiments, and thus there is no long-term trend or seasonal cycle. However, it has been suggested that waveguide characteristics might change in a warmer climate potentially favoring wave amplification (Mann et al., 2017, 2018). Although a recent review by Teng and Branstator (2019) concludes that changes in the diabatic heating are a more probable candidate for more high-amplitude circumglobal planetary wave events in the future, reported changes in the zonal mean circulation for example, from an overall slow-down (Coumou et al., 2015) or strengthening of the jet in certain latitudes (Xu et al., 2020) will likely affect the waveguiding ability of the zonal jet.

Impacts of heatwaves are strongly amplified with their increased persistence. While topography plays an essential role in determining the longitudinal structure of circumglobal wave patterns, its presence does not lead to a higher persistence of the high-amplitude events and might even lower their persistence in some cases (Figure S9 in Supporting Information S1). Hence, other components of the climate system not included in our idealized experiments may be responsible for an increased persistence of heatwaves associated with amplified wave patterns. Important mechanisms include land-surface feedbacks (Miralles et al., 2019; Schumacher et al., 2019; Teng et al., 2019) and diabatic heating maintaining upper level blocking anticyclones (Pfahl et al., 2015; Zschenderlein et al., 2020). Further work is required to improve our understanding of the relative importance of the different processes that drive amplification, phase-locking and maintenance of amplified waves. Models struggle with an accurate simulation of circumglobal wave patterns (e.g., Luo et al., 2021) and the magnitude of associated weather extremes. Understanding how the stationary components of the climate system (such as topography) interact with those components that might be affected by climatic change (such as land-atmosphere interactions and the circulation itself) is important for reducing model biases in order to improve our ability to predict and project changes in concurrent extreme events such as heatwaves in current and future climates.

Data Availability Statement

The ERA-Interim reanalysis (Dee et al., 2011) has been obtained from the ECMWF server (https://apps.ecmwf. int/datasets/data/interim-full-daily/levtype=pl/, last access: 27 September 2021). All analyses and visualizations were carried out using the Python packages NumPy (https://numpy.org/, last access: 21 June 2022), Matplot-lib (https://matplotlib.org/, last access: 21 June 2022), xarray (https://docs.xarray.dev/en/stable/, last access: 21 June 2022), and SciPy (https://scipy.org/, last access: 21 June 2022). The code to reproduce the analysis and the main figures of this paper can be found in GitHub (https://github.com/bernatj/paper_GRL_phase_locked_

Acknowledgments

The authors would like to thank the Center for Climate Systems Modeling (C2SM) at ETH Zurich for providing technical support and for maintaining the ICON model code repository. This research has been supported by the European Research Council (ERC) (Grant 847456). Kai Kornhuber was partially supported by the NSF project NSF AGS-1934358. Support from the Swiss National Science Foundation through projects PP00P2_170523 and PP00P2 198896 to D.D. is gratefully acknowledged. This work was supported by a grant from the Swiss National Supercomputing Centre (CSCS) under project ID s903. We also thank Dr. Volkmar Wirth and the anonymous reviewers for their time and insightful comments during the review of our paper.

circumglobal_heat_extremes.git). The daily mean model output is available in zenodo (https://doi.org/10.5281/ zenodo.7270361).

References

- Beverley, J. D., Woolnough, S. J., Baker, L. H., Johnson, S. J., Weisheimer, A., & O'Reilly, C. H. (2021). Dynamical mechanisms linking Indian monsoon precipitation and the circumglobal teleconnection. *Climate Dynamics*, 1(9–10), 3. https://doi.org/10.1007/s00382-021-05825-6
- Boers, N., Goswami, B., Rheinwalt, A., Bookhagen, B., Hoskins, B., & Kurths, J. (2019). Complex networks reveal global pattern of extreme-rainfall teleconnections. *Nature*, 566(7744), 373–377. https://doi.org/10.1038/s41586-018-0872-x
- Branstator, G. (2002). Circumglobal teleconnections, the jet stream waveguide, and the North Atlantic Oscillation. *Journal of Climate*, 15(14), 1893–1910. https://doi.org/10.1175/1520-0442(2002)015<1893:CTTJSW>2.0.CO;2
- Brayshaw, D. J., Hoskins, B., & Blackburn, M. (2009). The basic ingredients of the North Atlantic storm track. Part I: Land–sea contrast and orography. *Journal of the Atmospheric Sciences*, 66(9), 2539–2558. https://doi.org/10.1175/2009JAS3078.1
- Chang, E. K. M. (2009). Diabatic and orographic forcing of northern winter stationary waves and Storm tracks. Journal of Climate, 22(3), 670–688. https://doi.org/10.1175/2008JCL12403.1
- Charney, J. G., & Eliassen, A. (1949). A numerical method for predicting the perturbations of the middle latitude westerlies. *Tellus*, 1(2), 38–54. https://doi.org/10.3402/tellusa.v1i2.8500
- Cook, K. H., & Held, I. M. (1992). The stationary response to large-scale orography in a general circulation model and a linear model. *Journal of the Atmospheric Sciences*, 49(6), 525–539. https://doi.org/10.1175/1520-0469(1992)049<0525:TSRTLS>2.0.CO;2
- Coumou, D., Lehmann, J., & Beckmann, J. (2015). The weakening summer circulation in the Northern Hemisphere mid-latitudes. *Science*, 348(6232), 324–327. https://doi.org/10.1126/SCIENCE.1261768
- Coumou, D., Petoukhov, V., Rahmstorf, S., Petri, S., & Schellnhuber, H. J. (2014). Quasi-resonant circulation regimes and hemispheric synchronization of extreme weather in boreal summer. Proceedings of the National Academy of Sciences of the United States of America, 111(34), 12331–12336. https://doi.org/10.1073/pnas.1412797111
- Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137(656), 553–597. https://doi. org/10.1002/qi.828
- Di Capua, G., Sparrow, S., Kornhuber, K., Rousi, E., Osprey, S., Wallom, D., et al. (2021). Drivers behind the summer 2010 wave train leading to Russian heatwave and Pakistan flooding. npi Climate and Atmospheric Science. 4(1), 1–14. https://doi.org/10.1038/s41612-021-00211-9
- Ding, Q., & Wang, B. (2005). Circumglobal teleconnection in the Northern Hemisphere summer. Journal of Climate, 18(17), 3483–3505. https:// doi.org/10.1175/JCLI3473.1
- Domeisen, D. I. V., Martius, O., & Jiménez-Esteve, B. (2018). Rossby wave propagation into the Northern Hemisphere stratosphere: The role of zonal phase speed. *Geophysical Research Letters*, 45(4), 2064–2071. https://doi.org/10.1002/2017GL076886
- Domeisen, D. I. V., White, C. J., Afargan-Gerstman, H., Muñoz, Á. G., Janiga, M. A., Vitart, F., et al. (2022). Advances in the subseasonal prediction of extreme events: Relevant case studies across the globe. *Bulletin of the American Meteorological Society*, 103(6), E1473–E1501. https://doi.org/10.1175/BAMS-D-20-0221.1
- Drouard, M., Kornhuber, K., & Woollings, T. (2019). Disentangling dynamic contributions to summer 2018 anomalous weather over Europe. Geophysical Research Letters, 46(21), 12537–12546. https://doi.org/10.1029/2019GL084601
- Duchez, A., Frajka-Williams, E., Josey, S. A., Evans, D. G., Grist, J. P., Marsh, R., et al. (2016). Drivers of exceptionally cold North Atlantic Ocean temperatures and their link to the 2015 European heat wave. *Environmental Research Letters*, 11(7), 074004. https://doi. org/10.1088/1748-9326/11/7/074004
- Garfinkel, C. I., White, I., Gerber, E. P., Jucker, M., & Erez, M. (2020). The building blocks of Northern Hemisphere wintertime stationary waves. Journal of Climate, 33(13), 5611–5633. https://doi.org/10.1175/JCLI-D-19-0181.1
- Grotjahn, R. (2015). Dynamical meteorology Baroclinic instability. In G. R. North, J. Pyle, & F. Zhang (Eds.), Encyclopedia of atmospheric sciences (2nd ed., pp. 304–312). Academic Press. https://doi.org/10.1016/B978-0-12-382225-3.00076-1
- Harnik, N., Messori, G., Caballero, R., & Feldstein, S. B. (2016). The circumglobal North American wave pattern and its relation to cold events in eastern North America. *Geophysical Research Letters*, 43(20), 11015–11023. https://doi.org/10.1002/2016GL070760
- Harvey, B. J., Cook, P., Shaffrey, L. C., & Schiemann, R. (2020). The response of the Northern Hemisphere storm tracks and jet streams to climate change in the CMIP3, CMIP5, and CMIP6 climate models. *Journal of Geophysical Research: Atmospheres*, 125(23), e2020JD032701. https:// doi.org/10.1029/2020JD032701
- Hayashi, Y. (1979). A generalized method of resolving transient disturbances into standing and traveling waves by space-time spectral analysis. *Journal of the Atmospheric Sciences*, *36*(6), 1017–1029. https://doi.org/10.1175/1520-0469(1979)036<1017:agmort>2.0.co;2
- Held, I. M., & Suarez, M. J. (1994). A proposal for the intercomparison of the dynamical cores of atmospheric general circulation models. *Bulletin of the American Meteorological Society*, 75(10), 1825–1830. https://doi.org/10.1175/1520-0477(1994)075<1825:APFTIO>2.0.CO;2
- Held, I. M., Ting, M., & Wang, H. (2002). Northern winter stationary waves: Theory and modeling. *Journal of Climate*, 15(16), 2125–2144. https://doi.org/10.1175/1520-0442(2002)015<2125:NWSWTA>2.0.CO;2
- Hoegh-Guldberg, O., Jacob, D., Taylor, M., Bindi, M., Brown, S., Camilloni, I., et al. (2018). Impacts of 1.5 C global warming on natural and human systems. In V.Masson-Delmotte, P. Zhai, H.-O. Pörtner, D. Roberts, J. Skea, P. R. Shukla, et al. (Eds.), Global warming of 1.5°C. An IPCC special report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. Intergovernmental Panel on Climate Change.
- Huntingford, C., Jeffers, E. S., Bonsall, M. B., Christensen, H. M., Lees, T., & Yang, H. (2019). Machine learning and artificial intelligence to aid climate change research and preparedness. *Environmental Research Letters*, 14(12), 124007. https://doi.org/10.1088/1748-9326/AB4E55
- Jiménez-Esteve, B., & Domeisen, D. I. (2022). The role of atmospheric dynamics and large-scale topography in driving heatwaves. *Quarterly Journal of the Royal Meteorological Society*, 148(746), 2344–2367. https://doi.org/10.1002/qj.4306
- Kaspi, Y., & Schneider, T. (2011). Winter cold of eastern continental boundaries induced by warm ocean waters. *Nature*, 471(7340), 621–624. https://doi.org/10.1038/nature09924
- Koppe, C., Kovats, S., Jendritzky, G., & Menne, B. (2004). *Heat-waves: Risks and responses*. World Health Organization, Regional Office for Europe.
- Kornhuber, K., Coumou, D., Vogel, E., Lesk, C., Donges, J. F., Lehmann, J., & Horton, R. M. (2020). Amplified Rossby waves enhance risk of concurrent heatwaves in major breadbasket regions. *Nature Research*, 10(1), 48–53. https://doi.org/10.1038/s41558-019-0637-z

- Kornhuber, K., Osprey, S., Coumou, D., Petri, S., Petoukhov, V., Rahmstorf, S., & Gray, L. (2019). Extreme weather events in early summer 2018 connected by a recurrent hemispheric wave-7 pattern. *Environmental Research Letters*, 14(5), 054002. https://doi.org/10.1088/1748-9326/ ab13bf
- Kornhuber, K., Petoukhov, V., Karoly, D., Petri, S., Rahmstorf, S., & Coumou, D. (2017). Summertime planetary wave resonance in the northern and southern Hemispheres. *Journal of Climate*, 30(16), 6133–6150. https://doi.org/10.1175/JCLI-D-16-0703.1
- Kornhuber, K., Petoukhov, V., Petri, S., Rahmstorf, S., & Coumou, D. (2017). Evidence for wave resonance as a key mechanism for generating high-amplitude quasi-stationary waves in boreal summer. *Climate Dynamics*, 49(5–6), 1961–1979. https://doi.org/10.1007/s00382-016-3399-6 Kornhuber, K., & Tamarin-Brodsky, T. (2021). Future changes in Northern Hemisphere summer weather persistence linked to projected Arctic
- warming. Geophysical Research Letters, 48(4), e2020GL091603. https://doi.org/10.1029/2020GL091603
- Kosaka, Y., Nakamura, H., Watanabe, M., & Kimoto, M. (2009). Analysis on the dynamics of a wave-like teleconnection pattern along the Summertime Asian jet based on a reanalysis dataset and climate model simulations. *Journal of the Meteorological Society of Japan. Series II*, 87(3), 561–580. https://doi.org/10.2151/JMSJ.87.561
- Lau, W. K. M., & Kim, K.-M. (2012). The 2010 Pakistan flood and Russian heat wave: Teleconnection of hydrometeorological extremes. Journal of Hydrometeorology, 13(1), 392–403. https://doi.org/10.1175/JHM-D-11-016.1
- Lopez, H., West, R., Dong, S., Goni, G., Kirtman, B., Lee, S.-K., & Atlas, R. (2018). Early emergence of anthropogenically forced heat waves in the western United States and Great Lakes. *Nature Climate Change*, 8(5), 414–420. https://doi.org/10.1038/s41558-018-0116-y
- Lu, R.-Y., Oh, J.-H., & Kim, B.-J. (2002). A teleconnection pattern in upper-level meridional wind over the North African and Eurasian continent in summer. *Tellus A: Dynamic Meteorology and Oceanography*, 54(1), 44–55. https://doi.org/10.3402/tellusa.v54i1.12122
- Luo, F., Selten, F., Wehrli, K., Kornhuber, K., Le Sager, P., May, W., et al. (2021). Summertime circumglobal Rossby waves in climate models: Small biases in upper-level circulation create substantial biases in surface imprint. Weather and Climate Dynamics Discussions, 2021, 1–30. https://doi.org/10.5194/wcd-2021-48
- Lutsko, N. J., Baldwin, J. W., & Cronin, T. W. (2019). The impact of large-scale orography on Northern Hemisphere winter synoptic temperature variability. *Journal of Climate*, 32(18), 5799–5814. https://doi.org/10.1175/JCLI-D-19-0129.1
- Mann, M. E., Rahmstorf, S., Kornhuber, K., Steinman, B. A., Miller, S. K., & Coumou, D. (2017). Influence of anthropogenic climate change on planetary wave resonance and extreme weather events. *Scientific Reports*, 7(1), 7–12. https://doi.org/10.1038/srep45242
- Mann, M. E., Rahmstorf, S., Kornhuber, K., Steinman, B. A., Miller, S. K., Petri, S., & Coumou, D. (2018). Projected changes in persistent extreme summer weather events: The role of quasi-resonant amplification. *Science Advances*, 4(10), eaat3272. https://doi.org/10.1126/sciadv. aat3272
- Manola, I., Selten, F., de Vries, H., & Hazeleger, W. (2013). "Waveguidability" of idealized jets. *Journal of Geophysical Research: Atmospheres*, 118(18), 10432–10440. https://doi.org/10.1002/JGRD.50758
- Miralles, D. G., Gentine, P., Seneviratne, S. I., & Teuling, A. J. (2019). Land–atmospheric feedbacks during droughts and heatwaves: State of the science and current challenges. Annals of the New York Academy of Sciences, 1436(1), 19–35. https://doi.org/10.1111/nyas.13912
- Narinesingh, V., Booth, J. F., Clark, S. K., & Ming, Y. (2020). Atmospheric blocking in an aquaplanet and the impact of orography. *Weather and Climate Dynamics*, 1(2), 293–311. https://doi.org/10.5194/wcd-1-293-2020
- Osman, M. B., Coats, S., Das, S. B., McConnell, J. R., & Chellman, N. (2021). North Atlantic jet stream projections in the context of the past 1,250 years. Proceedings of the National Academy of Sciences, 118(38), e2104105118. https://doi.org/10.1073/PNAS.2104105118
- Perkins, S., Alexander, L., & Nairn, J. (2012). Increasing frequency, intensity and duration of observed global heatwaves and warm spells. *Geophysical Research Letters*, 39(20), 2012GL053361. https://doi.org/10.1029/2012gl053361
- Perkins-Kirkpatrick, S. E., & Lewis, S. C. (2020). Increasing trends in regional heatwaves. *Nature Communications*, 11(1), 3357. https://doi. org/10.1038/s41467-020-16970-7
- Petoukhov, V., Petri, S., Kornhuber, K., Thonicke, K., Coumou, D., & Schellnhuber, H. J. (2018). Alberta wildfire 2016: Apt contribution from anomalous planetary wave dynamics. *Scientific Reports*, 8(1), 1–10. https://doi.org/10.1038/s41598-018-30812-z
- Petoukhov, V., Petri, S., Rahmstorf, S., Coumou, D., Kornhuber, K., & Schellnhuber, H. J. (2016). Role of quasiresonant planetary wave dynamics in recent boreal spring-to-autumn extreme events. *Proceedings of the National Academy of Sciences*, 113(25), 6862–6867. https://doi. org/10.1073/PNAS.1606300113
- Petoukhov, V., Rahmstorf, S., Petri, S., & Schellnhuber, H. J. (2013). Quasiresonant amplification of planetary waves and recent Northern Hemisphere weather extremes. Proceedings of the National Academy of Sciences of the United States of America, 110(14), 5336–5341. https://doi. org/10.1073/pnas.1222000110
- Pfahl, S., Schwierz, C., Croci-Maspoli, M., Grams, C. M., & Wernli, H. (2015). Importance of latent heat release in ascending air streams for atmospheric blocking. *Nature Geoscience*, 8(8), 610–614. https://doi.org/10.1038/ngeo2487
- Philip, S. Y., Kew, S. F., van Oldenborgh, G. J., Anslow, F. S., Seneviratne, S. I., Vautard, R., et al. (2021). Rapid attribution analysis of the extraordinary heatwave on the Pacific coast of the US and Canada June 2021. *Earth System Dynamics Discussions*, 2021, 1–34. https://doi. org/10.5194/esd-2021-90
- Randel, W. J., & Held, I. M. (1991). Phase speed spectra of transient eddy fluxes and critical layer absorption. *Journal of the Atmospheric Sciences*, 48(5), 688–697. https://doi.org/10.1175/1520-0469(1991)048<0688:PSSOTE>2.0.CO;2
- Riboldi, J., Rousi, E., D'Andrea, F., Rivière, G., & Lott, F. (2021). Circumglobal Rossby wave patterns during boreal winter highlighted by wavenumber/phase speed spectral analysis. Weather and Climate Dynamics Discussions, 2021, 1–26. https://doi.org/10.5194/wcd-2021-59
- Saulière, J., Brayshaw, D. J., Hoskins, B., & Blackburn, M. (2012). Further investigation of the impact of idealized continents and SST distributions on the Northern Hemisphere storm tracks. *Journal of the Atmospheric Sciences*, 69(3), 840–856. https://doi.org/10.1175/JAS-D-11-0113.1
- Schumacher, D. L., Keune, J., van Heerwaarden, C. C., Vilà-Guerau de Arellano, J., Teuling, A. J., & Miralles, D. G. (2019). Amplification of mega-heatwaves through heat torrents fuelled by upwind drought. *Nature Geoscience*, 12(9), 712–717. https://doi.org/10.1038/ s41561-019-0431-6
- Singh, D., Swain, D. L., Mankin, J. S., Horton, D. E., Thomas, L. N., Rajaratnam, B., & Diffenbaugh, N. S. (2016). Recent amplification of the North American winter temperature dipole. *Journal of Geophysical Research: Atmospheres*, 121(17), 9911–9928. https://doi. org/10.1002/2016JD025116
- Teng, H., & Branstator, G. (2019). Amplification of waveguide teleconnections in the boreal summer. Springer, 5(4), 421–432. https://doi.org/10.1007/s40641-019-00150-x
- Teng, H., Branstator, G., Tawfik, A., & Callaghan, P. (2019). Circumglobal response to prescribed soil moisture over North America. Journal of Climate, 32(14), 4525–4545. https://doi.org/10.1175/JCLI-D-18-0823.1
- Teng, H., Branstator, G., Wang, H., Meehl, G. A., & Washington, W. M. (2013). Probability of US heat waves affected by a subseasonal planetary wave pattern. *Nature Geoscience*, 6(12), 1056–1061. https://doi.org/10.1038/ngeo1988

- Valdes, P. J., & Hoskins, B. J. (1991). Nonlinear orographically forced planetary waves. *Journal of the Atmospheric Sciences*, 48(18), 2089–2106. https://doi.org/10.1175/1520-0469(1991)048<2089:NOFPW>2.0.CO;2
- Vogel, M. M., Zscheischler, J., Wartenburger, R., Dee, D., & Seneviratne, S. I. (2019). Concurrent 2018 hot extremes across Northern Hemisphere due to human-induced climate change. *Earth's Future*, 7(7), 692–703. https://doi.org/10.1029/2019EF001189
- Watanabe, M., Shiogama, H., Imada, Y., Mori, M., Ishii, M., & Kimoto, M. (2013). Event attribution of the August 2010 Russian heat wave. SOLA, 9(0), 65–68. https://doi.org/10.2151/sola.2013-015
- Wehrli, K., Hauser, M., & Seneviratne, S. I. (2020). Storylines of the 2018 Northern Hemisphere heatwave at pre-industrial and higher global warming levels. *Earth System Dynamics*, 11(4), 855–873. https://doi.org/10.5194/ESD-11-855-2020
- White, R. H., Battisti, D. S., & Roe, G. H. (2017). Mongolian mountains matter most: Impacts of the latitude and height of Asian orography on Pacific wintertime atmospheric circulation. *Journal of Climate*, *30*(11), 4065–4082. https://doi.org/10.1175/JCLI-D-16-0401.1
- White, R. H., Kornhuber, K., Martius, O., & Wirth, V. (2021). From atmospheric waves to heatwaves: A waveguide perspective for understanding and predicting concurrent, persistent and extreme extratropical weather. *Bulletin of the American Meteorological Society*, 103(3), 1–35. https:// doi.org/10.1175/BAMS-D-21-0170.1
- White, R. H., Kornhuber, K., Martius, O., & Wirth, V. (2022). From atmospheric waves to heatwaves: A waveguide perspective for understanding and predicting concurrent, persistent, and extreme extratropical weather. *Bulletin of the American Meteorological Society*, 103(3), E923–E935. https://doi.org/10.1175/BAMS-D-21-0170.1
- Wirth, V. (2020). Waveguidability of idealized midlatitude jets and the limitations of ray tracing theory. Weather and Climate Dynamics, 1(1), 111–125. https://doi.org/10.5194/wcd-1-111-2020
- Wirth, V., & Polster, C. (2021). The problem of diagnosing jet waveguidability in the presence of large-amplitude eddies. Journal of the Atmospheric Sciences, 78(10), 3137–3151. https://doi.org/10.1175/JAS-D-20-0292.1
- Wirth, V., Riemer, M., Chang, E. K. M., & Martius, O. (2018). Rossby wave packets on the midlatitude waveguide—A review. *Monthly Weather Review*, 146(7), 1965–2001. https://doi.org/10.1175/MWR-D-16-0483.1
- Xu, P., Wang, L., Liu, Y., Chen, W., & Huang, P. (2020). The record-breaking heat wave of June 2019 in Central Europe. Atmospheric Science Letters, 21(4), e964. https://doi.org/10.1002/asl.964
- Yaghmaei, N. (2020). Human cost of disasters: An overview of the last 20 years, 2000–2019. UN Office for Disaster Risk Reduction. Retrieved from https://www.undrr.org/publication/human-cost-disasters-overview-last-20-years-2000-2019
- Zängl, G., Reinert, D., Rípodas, P., & Baldauf, M. (2015). The ICON (ICOsahedral Non-hydrostatic) modelling framework of DWD and MPI-M: Description of the non-hydrostatic dynamical core. *Quarterly Journal of the Royal Meteorological Society*, 141(687), 563–579. https://doi. org/10.1002/qj.2378
- Zschenderlein, P., Pfahl, S., Wernli, H., & Fink, A. H. (2020). A Lagrangian analysis of upper-tropospheric anticyclones associated with heat waves in Europe. *Weather and Climate Dynamics*, 1(1), 191–206. https://doi.org/10.5194/wcd-1-191-2020